Author's response to reviews

Title: Self-induced Elizabethkingia meningosepticum endophthalmitis

Authors:

Paul P Connell (drpaulconnell@gmail.com)
Sanj Wicksameringue (Sanj.Wickremasinghe@eyeandear.org.au)
Uma Devi (u.devi@svhm.org.au)
Mary Jo Waters (mary.waters@svhm.org.au)
Penny Allen (gillall@ozemail.com.au)

Version: 2 Date: 27 April 2010

Author's response to reviews: see over
Cover Letter

Dr Paul Connell
Consultant Ophthalmologist
Royal Victorian Eye and Ear Hospital
Melbourne Australia
3205

To the Editor,

Please find attached our revised submission entitled “Self-induced Elizabethkingia meningosepticum endophthalmitis” which we are pleased to submit for consideration in the Journal of Medical Case Reports. We believe that the manuscript presents a rare and unusual case of endophthalmitis. The aetiological nature of the insult coupled with the age of the patient at presentation have, to our knowledge, not previously been reported. The case report highlights the importance of close ocular and medical history especially in the context of the associated microbiological diagnosis. The new revision contains the changes requested- specifically the ethnicity of the patient and a conclusion section in main text.

Many thanks for considering this submission for publication.

With best wishes,
Paul Connell